Author’s response to reviews

Title: Challenges and opportunities in the continuity of care for hypertension: a mixed-methods study embedded in a primary health care intervention in Tajikistan

Authors:

Adanna Chukwuma (achukwuma@worldbank.org)
Estelle Gong (estelletgong@gmail.com)
Mutriba Latypova (mlatypova@worldbank.org)
Nicole Fraser-Hurt (nicolefraser@btinternet.com)

Version: 2 Date: 18 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you for the favorable response and speedy turnaround time for this submission.

Following your email, we have made the following changes:

1) Noted that the funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript [Lines 624-7]
2) Renamed Figure 4 appropriately and checked that the citation of Figures in the text is appropriate.
3) Removed colored text in the main manuscript.

Thank you,

Adanna Chukwuma